Author’s response to reviews

Title: Coexistence of Primary adenocarcinoma of the lung and Tsukamurella pneumonia: a case report and review of the literature.

Authors:

Vinicio A de Jesus Perez (vdejesus@stanford.edu)
Jeffrey Swigris (swigrisi@nij.org)
Stephen Ruoss (ruoss@stanford.edu)

Version: 3 Date: 15 January 2008

Author’s response to reviews:

Dear Editor-in-Chief,

I am sending you the latest revised version of our manuscript. We have addressed the issues raised by the reviewers on the first revision of the manuscript and sincerely hope that this version meets the strict quality standards of your publication. The following comments were addressed:

1. The clinical condition of the patient at the end of the therapy, susceptibility data, and the duration of the therapy has been stated.
2. Modifications requested by referee 2 have been applied to the text and images.
3. We agree that our recommendations for antibiotic therapy are limited by lack of controlled studies and cannot rule out the possibility of using other antibiotics either as single therapy or in combination to treat Tsukamurella pneumonia. Our recommendations for management of this condition are based on our experience with this patient and may require revision when controlled studies looking at these questions become available.

Thanks again for your continued interest in this work. I hope to hear from you shortly regarding the decision to publish this case.

Sincerely,

Vinicio A. de Jesus Perez, MD